Week 18 starting 01 May 06
SM18 Operation Meeting (Tuesdays) 2 May 06
Bench status  on Monday  01-05-2006 at: 5/1/2006 11:03:24 AM  
	12
	A1
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	299.47

	3
	A2
	MBAL1387
	ICS 5 Traveller Last
	299.65

	5
	B1
	SSS182
	PT 9 CDAP HV
	1.90

	1
	B2
	SSS232
	ICS 3 Traveller after warm test
	299.65

	2
	C1
	MBBL2180
	CR 1 Leak Test
	298.04

	8
	C2
	MBAR1323
	PT 1 IAP @ cold
	1.91

	11
	D1
	SSS636
	Cryo problem
	298.31

	9
	D2
	SSS609
	PT 1 IAP @ cold
	2.35

	7
	E1
	MBAR2317
	PT 14 Magnet MAPS
	300.82

	4
	E2
	MBAR2236
	PT 12 Warm Up
	260.22

	10
	F1
	MBBR1344
	Prep 5 Cool Down
	62.88

	6
	F2
	MBAR1297
	PT 6.1 Training 1
	7.81


  
Supervisor: stephane/        Eq support : 16-3602 :
14 Magnet Tests completed :  8 Dipoles (),    5 Arc-SSS  ( 182 to be  REPEATed)





     :

 1 500-series ( 503 )  
:






 0  600-series (  )
· { D1:636 Leak problems: waited whole weekend 22-23April till Mon 24th ..no luck  Now disconnected since Friday 28th but could not be moved }waiting for 635 on D1

· ROCLA ACCIDENT LAST Wednesday evening 26 -4-06: blocked usage till Safety green light (24 hrs) + not able to access to A cluster at all ;
Final removal from front of A2 on Friday 28th evening; we were blocked for removals also for cluster E,F( & D) because ROCLA could not be used neither from outside big door Jura end ..due to outside metal floor goodness ?
General Points :
· C Cluster & Output !! no dipole magnets done yet  after 1 week of changeover!!  
Need Arc-SSS buffer of  ‘to test’ magnets as well as for  ‘non-MM’  SSSS 

· Situation from Cryogenics:  J-P Lamboy or ACR Rep.
· News /Situation from (SSS, SSSS)  :  J Halik
· Situation SSSS  :OP issues :
· A.O.B.
	

	Magnets under Test from 23-04-2006 to 29-04-2006 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBAR2323 
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	21-04-2006 
	32 h.
	29 h.
	23 h.
	15 h.
	15 h.
	26-04-2006 
	114 h.
	Stripping
	
	

	2
	TBA2
	MBAR1272 
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	19-04-2006 
	34 h.
	21 h.
	56 h.
	22 h.
	10 h.
	25-04-2006 
	143 h.
	Stripping
	
	

	3
	TBB1
	SSS182 
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	25-04-2006 
	25 h.
	21 h.
	-
	-
	-
	29-04-2006 
	83 h.
	Not Found or Thermal cycle
	
	

	4
	TBB1
	SSS322 
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	21-04-2006 
	30 h.
	16 h.
	15 h.
	13 h.
	19 h.
	25-04-2006 
	93 h.
	Stripping
	
	

	5
	TBB2
	SSS247 
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	22-04-2006 
	29 h.
	12 h.
	32 h.
	22 h.
	10 h.
	26-04-2006 
	105 h.
	Stripping
	
	

	6
	TBC1
	SSS291 
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	20-04-2006 
	30 h.
	19 h.
	22 h.
	7 h.
	11 h.
	24-04-2006 
	89 h.
	Stripping
	
	

	7
	TBC2
	SSS246 
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	20-04-2006 
	29 h.
	25 h.
	33 h.
	8 h.
	43 h.
	26-04-2006 
	138 h.
	Stripping
	
	

	8
	TBD2
	SSS503 
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	20-04-2006 
	49 h.
	15 h.
	28 h.
	7 h.
	22 h.
	25-04-2006 
	121 h.
	Stripping
	
	

	9
	TBE1
	MBBL1236 
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	18-04-2006 
	32 h.
	22 h.
	42 h.
	13 h.
	11 h.
	23-04-2006 
	120 h.
	Stand By
	
	

	10
	TBE1
	MBBL2180 
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	23-04-2006 
	-
	-
	-
	-
	-
	25-04-2006 
	37 h.
	Not Found or Thermal cycle
	
	

	11
	TBE2
	MBAR2236 
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	24-04-2006 
	34 h.
	26 h.
	32 h.
	16 h.
	0 h.
	29-04-2006 
	108 h.
	Not Found or Thermal cycle
	
	

	12
	TBE2
	MBBR1313 
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	19-04-2006 
	24 h.
	19 h.
	55 h.
	14 h.
	9 h.
	24-04-2006 
	121 h.
	Stripping
	
	

	13
	TBF1
	MBBR2298 
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	22-04-2006 
	29 h.
	22 h.
	28 h.
	15 h.
	49 h.
	28-04-2006 
	143 h.
	Stripping
	
	

	14
	TBF2
	MBAR1336 
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	20-04-2006 
	26 h.
	25 h.
	32 h.
	5 h.
	3 h.
	24-04-2006 
	91 h.
	Stripping
	
	

	15
	TBF2
	MBBR2283 
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	24-04-2006 
	28 h.
	19 h.
	21 h.
	15 h.
	6 h.
	28-04-2006 
	89 h.
	Stripping
	
	

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAR2323
	12074.3 A.
(D2_L)
	12853 A.
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAR1272
	12729 A.
(D2_L)
	12852.2 A.
(D1_L)
	12852 A.
[image: image33.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	SSS182
	10786.3 A.
(QE_34)
	11754.8 A.
(QE_34)
	12070.6 A.
(QE_34)
	12211.3 A.
(QE_12)
	12298 A.
(QE_34)
	12432.4 A.
(QE_34)
	12422.9 A.
(QE_34)
	12553.3 A.
(QE_12)
	12642.5 A.
(QE_12)
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	-
	-
	-

	4
	SSS322
	10950 A.
(No Data)
	11983 A.
(QI_12)
	12850 A.
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	SSS247
	12525.8 A.
(QE_12)
	12853 A.
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS291
	11956.2 A.
(QE-12)
	12817 A.
(QE_12)
	12850 A.
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS246
	12225.7 A.
(QI-12)
	12416.7 A.
(QE_34)
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	SSS503
	11854.3 A.
(QE_34)
	12277.7 A.
(QE_12)
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBL1236
	10890.1 A.
(D1_L)
	11941 A.
(D2_U)
	12134 A.
(D1_L)
	12296.8 A.
(D1_U)
	12467.2 A.
(D1_L)
	12573.9 A.
(D2_U)
	12720.8 A.
(D1_U)
	12652 A.
(D1_L)
	12702 A.
(D1-L)
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	-
	-
	-

	10
	MBBL2180
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAR2236
	11881.3 A.
(D2_L)
	12403.1 A.
(D1_U)
	12247.9 A.
(D1_U)
	11843.7 A.
(D1_U)
	11403.3 A.
(D1_U)
	11069.1 A.
(D1_U)
	11492.3 A.
(D1_U)
	12614 A.
(D2_U)
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	-
	-
	-
	-

	12
	MBBR1313
	12272 A.
(D1-U)
	12633.4 A.
(D2-U)
	12708.4 A.
(D2-L)
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	-
	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBBR2298
	10422.8 A.
(D1-U)
	12263 A.
(D1 - U)
	11030 A.
(D2-U)
	12574.8 A.
(D1_L)
	12669.9 A.
(D2_L)
	12728.6 A.
(D1-U)
	12829.5 A.
(D1_U)
	12850 A.
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	-
	-
	-
	-
	-

	14
	MBAR1336
	11992 A.
(D2-U)
	12289.4 A.
(D1-L)
	12404 A.
(D1_U)
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	-
	-
	-
	-
	-
	-
	-
	-
	-

	15
	MBBR2283
	11448.8 A.
(D2_U)
	12182.5 A.
(D1_L)
	12334 A.
(D2_L)
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	-
	-
	-
	-
	-
	-
	-
	-
	-
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No  600 series at all this week;  ROCLA Accident effect for removal on 2298 ( on F  cluster) ; To dipole Conversion for C cluster on SSS246  
SSS182 Thermal cycle;  2180 on E was removed because HV failure after Connection( before was ok!); Now 2180 on C cluster & again a problem  & had to be warmed up & disconnected & reconnected & re-cooled !
FOR APRIL 06  SM18 TESTS OUTPUT 

[image: image48.emf]0

5

10

15

20

25

30

35

40

Dipole SSS 600SSS

Predicted

Actual

 
ROCLA ACCIDENT STORY EMAILS

From: Ingo Ruehl
Sent: Fri 28/04/2006 17:39
To: Sylvain Weisz; Lyn Evans; Paolo Ciriani; Philippe Lebrun; Lucio Rossi; Alain Poncet; Louis Walckiers; Andrzej Siemko; Vinod Chohan
Cc: Claude Hauviller; Ralf Trant; Laurent Colly; Serge Pelletier; Oliver Boettcher
Subject: RE: Accident ROCLA
Bonsoir, 

A short update: 

The accident zone in SM18 was cleared up this afternoon without any incidents. 

The driving mechanism of the damaged ROCLA 3 (CH545) is operational. The main concern remains the replacement of the lifting mechanism which may have to include the gear box according to the assessment of ROCLA's mechanical engineer who was on site all day. A more detailed analyze can be done once the disassembly works start on Tuesday next week. We requested a support team from the manufacturer for the repair works. Nevertheless, we estimate a minimum of two weeks to bring the vehicle back in service.
The fine adjustment of the newly installed cable on ROCLA 1 (CH500) is still under way. The reassembling will be finished tonight so that the load tests can be carried out on Tuesday morning (in agreement with the users in SM18).
Regards, 
Ingo Rühl / TS-IC 

-----Original Message----- 
From: Sylvain Weisz 
Sent: Thursday, April 27, 2006 7:33 PM 
To: Lyn Evans; Paolo Ciriani; Philippe Lebrun; Lucio Rossi; Alain Poncet; Louis Walckiers; Andrzej Siemko; Vinod Chohan 
Cc: Claude Hauviller; Ralf Trant; Laurent Colly; Serge Pelletier; Oliver Boettcher; Ingo Ruehl 
Subject: RE: Accident ROCLA 

Bonsoir à tous, 
     Vous trouverez ci-dessous la conclusion de SC-GS (Ralf Trant et Laurent Colly) en ce qui concerne la remise en route des véhicules ROCLA. Pour le court terme, cela veut dire:
1) que le ROCLA2, qui est en configuration MB, est de nouveau en service; 
2) que l'on remplace les câbles du ROCLA1, qui est en configuration SSS: le véhicule sera prêt pour un test en charge demain en début d'après-midi et pourra ensuite être mis en service;
3) que le ROCLA3 peut être déchargé et enlevé une fois la procédure établie et validée par SC-GS. Des représentants de ROCLA viendrons dans la matinée pour faire leur propre analyse de l'accident et pour évaluer les réparations à entreprendre sur ce véhicule. La procédure d'évacuation de l'aimant et du ROCLA3 sera disponible demain dans la matinée: après validation par SC-GS, cette évacuation se fera dans l'après-midi.
                                                Bonne soirée, 
                                                        Sylvain 

Par mesure de sécurité, suite à l'incident survenu mercredi soir sur le véhicule de transport d'aimants ROCLA n° CH-545 au SM18, les 2 autres véhicules ROCLA ont été consignés. 
En regard des résultats de l'examen réalisé par M. VERDUYN de la Société APAVE sur les 3 véhicules, et suite à la réunion SC/GS - TS/IC/HM tenue cet après-midi, les dispositions suivantes ont été retenues: 
- compte tenu de l'absence de fil cassé ou de détérioration sur les 4 câbles de levage du ROCLA2 (CH-501), ce véhicule peut immédiatement être remis en service. Cette autorisation d'utilisation est soumise aux dispositions d'inspection renforcée des câbles décrites plus bas. 
- l'examen a démontré la nécessité de remplacer les 4 câbles de levage du ROCLA1 (CH-500). Leur remplacement sera suivi d'une réception avant remise en service. En cas d'autorisation de remise en service à l'issue de cette réception, les dispositions d'inspection renforcée des câbles seront également appliquées. 
- Rocla3: les analyses de SC/GS sont terminées. Le véhicule est laissé à la disposition de TS/IC/HM. 
NOTA: 
- les câbles détériorés et remplacés seront conservés afin de permettre leur analyse ultérieure. 
- la procédure d'évacuation de l'aimant endommagé et du Rocla3 devra être approuvée par SC/GS 
Disposition d'inspection renforcée des câbles des véhicules Rocla: 
- périodicité: 2 semaines 
En fonction du résultat des inspections ou/et de l'intensité d'utilisation des véhicules, la périodicité pourra être aménagée. 
- Un descriptif précis sera défini la semaine prochaine (procédure, critères de dépose des câbles). 
Laurent Colly - SC/GS 

-----Original Message----- 
From: Sylvain Weisz 
Sent: jeudi, 27. avril 2006 08:22 
To: Lyn Evans; Paolo Ciriani; Philippe Lebrun; Lucio Rossi; Alain Poncet; Louis Walckiers 
Cc: Claude Hauviller 
Subject: Accident ROCLA 

        Nous avons eu un accident avec le ROCLA Nr 3 au SM18 hier soir: un des 2 câbles porteurs s'est rompu sous la charge d'un dipole. Il n'y a pas de blessé, l'aimant est tombé ~1m sur une extrémité. Le système est actuellement en sécurité et nous cherchons l'origine de l'usure du câble. Les autres ROCLAs sont consignés jusqu'à ce que nous en sachions plus. Je vous tiens au courant dès que j'ai des informations plus précises.
                                                  Sylvain 

	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday_22_A
	TBE2
	MB2236 : TT 821 temperature sensor reading error. 
	MTM
	
	NC opened.

	Tuesday_25_N
	TBE1
	MB2317 : Error in temperature sensor reading in IAP at warm-Prep 2.1 and 2.2. (shows 80K instead of 300K).
	MTM
	
	Maryline advised to proceed to further tests. NC opened in CDW1.

	Tuesday_25_N
	TBD1
	SSS636: Cryo leak problem not solved. 
	CRI
	
	The SSS636 is being warmed up and then will be removed for repair.

	Wednesday_26_A
	TBA1
	Accident of the MBB2252: the magnet failed down from ROCLA due to arm-thread cable broken. 
	
	
	Test coordinator and Nick informed 

ICS responsible has asked firemen to come.
Actions taken by A. Siemko due to Rocla accident:

- No intervention on TBA1 and TBA2

- Forbidden access to prohibited zone around Rocla

- Guardians will frequently survey the area

- All Roclas operation stopped till further notice

- OP should put warning messages and transmit the information.

	Wednesday_26_N
	TBB2
	SSS234: Prep2.2C could not be completed (Clamp connection CD ). LF data didn't save, even though LF gets armed and gets triggered ( by 1A current ).

During check application after switching on DA switches, the following error message was observed " could not find USR_open.vi ". 
Check application didn’t complete in TEMA. 
	AB/CO
	
	OP team tried to reset DA switches after exiting web view from TEMA and all programs from sunMTA 
Eq. support contacted Thursday morning

	Wednesday_26_N
	TBE2
	MB2236: HV sequence failed at Cryogenic heater Vs Ground. 

	MTM
	
	Eq. Support contacted Thursday morning.

Equip. support advised to skip the step and continue with further tests.
NC opened.

	Thursday_27_A
	TBB1
	SSS182: HF DAQ did nor arm while launching PT5.3.
	AB/PO
	
	Problem with on/stand-by power supply.

	Thursday_27_A
	TBE2
	MB2236 : During PT8 (ramping up to 12000A) a quench occurred
	MTM
	
	TC has been contacted and advised to repeat the same test. Repeated test was successful.

	Thursday_27_N
	TBE2
	MB2236: Next training (warm up quench) attempted after HV. Magnet showed detraining behavior 12247A. TC contacted for further course. He suggested to train the magnet till ultimate. 
	MTM
	
	Detraining is continuing. Magnet will undergo a thermal cycle.

	Thursday_27_N
	TBA2
	MB1387 : For safety reason (ROCLA accident)TC instructed to stop work and put the magnet on 4.2K stand by till Saturday 9 AM and ask for further instruction
	
	
	

	Thursday_27_N
	TBE1
	MB2137 :PT1 on TBE1 shows error in D1-D2. 
	MTM
	
	Potaim cards not adjusted as training is on in E2. Therefore magnet in E1 is kept at cryo ok 4.2K.

	Thursday_27_N
	TBB1
	TEMA showing error message 'Magnet Absent in the Active Bench'. Not possible to launch PT 6.3. OP team tried to re-start TEMA and revalidate the scanned data of magnet through Mobic, but no success.


	IT
	HWREC
	Maryline contacted. She asked to put off HWREC and then enter the magnet names of B1 and B2 manually and proceed.

	Friday_28_A
	TBD2
	SSS609: Permanent trigger on Vclamp/Vsupra int/Ext
	MTM
	
	Maryline contacted.

	Friday_28_A
	TBB1
	SSS182: PLCSE communication lost during the training test.
	AB/CO
	
	PLCSE reset by switching off/on.

	Friday_28_N
	TBA1
	New magnet cannot be installed because big bellow and Y line are missing.
	ICS
	
	TC asked ICS Sunday to put another dipole with below. 

TC contacted Halik. A.Siemko has the key to open ICS store and get a new bellow

	Friday_28_N
	TBA2
	MB3187: PLCSE fault: Powering of the magnet could not be launched.
	MTM
	
	

	Friday_28_N
	TBC1
	MB2180: During QH test at 1.5 kA, trigger on the Vclamp Vtaps. Current not ramping, quench heaters getting discharged and files are generated.
	ICS
	
	Maryline informed. With TC they asked perform HV test final and warm up the magnet and check the connection in the M3 line. 

Magnet re open Sunday afternoon. Connection re-done and warm test re-started.

Training rules as for the first run applied but warm test results in the folder _3.

In fact problem with power supply parameter.

	Friday_28_N
	TBC2
	MB1323: Voltage Tap error in Prep2.1.
	ICS
	
	PLCSE blocked & Vtaps connectors on the top of CFB were inversed. Solved by equip support?

	Saturday_29_M
	
	Cryo OK lost for 4 benches at 1.9 K (A2, B2, C1, E1)
	ACR
	
	ACR team has no explanation.

	Saturday_29_M
	TBD2
	SSS609: SSS connection. Paper traveler missing
	MTM
	
	Who has it ?

 ICS, MTM mechanics? 

	Saturday_29_A
	TBB2
	SSS234: Corrector circuits GH and IJ not possible to power.
	MTM
	
	Fast stop was pushed.

	Saturday_29_A
	TBC1
	MB2180: Final HV test before warm up.  Test failed dipole quench-heater vs Ground.

HV Rack MY92:- Rack seems faulty. Tried 3 times on TBB2. PLCSE communication error.
	MTM
	
	Maryline contacted. She suggested to re-run and skip the failing steps. On 2nd run with different rack HV successful.

	Sunday_30_M
	TBB1
	SSS182: 1st training = 12046 A, then 2 wrong PT8 (quench at less than 12000 & not detected + PLSCE problem).
PTC6.4 = 12427 A and good analysis; before next PT8, V-power QH 4 discharged by its own
	MTM
	
	TC asked to redo a training quench and then PT8.

TC informed A. Siemko. He asked to perform 9 quenches or reach ultimate.

	Monday_01_M
	TBF2
	MB1297: PT6.1 - TEMA stopped executing the step RAMP to 12850 A (training quench) when switching to 10 A/s.
	MTM
	
	Shift leader executed the ramp manually. 

	Monday_01_M
	TBD2
	SSS609: Quench while going to Standby.
The magnet ramped to 50A but quenched again during the ramp-down to "OFF".
	MTM
	
	A.Siemko tried changing the threshold of MQY1&2_E-I. 
He suggested verifying the compensation and measuring QH resistances again to see if all is OK before continuing.

	Monday_01_M
	TBC2
	MB1323: During Slow Power Abort Test, Power Converter is not going to standby.
	MTM
	
	Guy contacted. Same problem as in C1. Internal parameter in the power supply not correct. Corrected by AB/PO support.

	Monday_01_A
	TBA2
	MB2338: Wrong magnet name on the code bar which could not be validated by the MOBIC
	ICS
	
	NC opened by OP team
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